Author's response to reviews

Title: Peer role-play and standardised patients in communication training: a comparative study on the student perspective on acceptability, realism, and perceived effect

Authors:

Hans M Bosse (hans.martin.bosse@med.uni-heidelberg.de)
Martin Nickel (martin.nickel@yahoo.de)
Sören Huwendiek (soeren.huwendiek@med.uni-heidelberg.de)
Jana Jünger (jana.juenger@med.uni-heidelberg.de)
Jobst H Schultz (jobst-hendrik.schultz@med.uni-heidelberg.de)
Christoph Nikendei (christoph.nikendei@med.uni-heidelberg.de)

Version: 9 Date: 26 March 2010

Author's response to reviews:

Covering letter
25/03/2010

Dear Editorial Production Team,

We have made three minor changes in the manuscript.

In the abstract two numbers were incorrect:
- p. 2, Results section, line 2: “RP 5.60”
- p. 2, Results, section line 4: “SP 5.50”.

In the Discussion section (p 14), paragraph ‘Acceptance’, line 3 a hyphen was removed: “peer role-play”

Thank you for your comprehensive editorial work. We are looking forward to the publication now.

Kind regards

Hans Martin Bosse

Dr. med. Hans Martin Bosse, MME

Specialist registrar

Outpatient Department, Medical Education
University Clinic Heidelberg
Centre of Child and Adolescent Medicine
INF 430
69120 Heidelberg
Germany
+49 - 6221 - 5639320